Author's response to reviews

Title: Left atrial volume index and aortic stiffness index in adult hemodialyzed patients - link between compliance and pressure mediated by endothelium dysfunction; a cross-sectional study

Authors:

Tomasz Zapolski (zapolia@wp.pl)
Andrzej Wysokiński (a.wysokinski@am.lublin.pl)
Andrzej Książek (ksan@interia.pl)
Andrzej Jaroszyński (jaroszynskiaj@interia.pl)

Version: 4 Date: 13 October 2012

Author's response to reviews: see over
Dear Marinette Lacson

Thank you for kind consideration for publication of our manuscript (1272997415747231) in the BioMed Central Cardiovascular Disorders. Accordingly to suggestions enclosed in Reviewer 2 commentary we revised the manuscript and we made some changes as follows (red font in manuscript):

Reviewer 2

We have performed an analysis of reproducibility of ASI and LAVI measurements and we described this in “Reproducibility” chapter.

We have chose the LAVI for analysis because LAVI is a new index of echocardiography and is compared to the “glycated haemoglobin of diabetes mellitus” as it is reflection of long-standing hemodynamic condition in different clinical settings.

We hope that our changes have improved quality of presented manuscript and thereby you will be able to consider this version of paper for publication in the BioMed Central Cardiovascular Disorders.

With compliments -
Tomasz Zapolski, Andrzej Wysokiński, Andrzej Książek, Andrzej Jaroszyński